Author's response to reviews

Title: A comparison of photographic, replication and direct clinical examination methods for detecting developmental defects of enamel

Authors:

Ali Golkari (aligolkari@yahoo.com)
Aira Sabokseir (aira_sabokseir@yahoo.com)
Hamid-Reza Pakshir (yadgari1@yahoo.com)
M. Christopher Dean (chris.dean@ucl.ac.uk)
Aubrey Sheiham (a.sheiham@ucl.ac.uk)
Richard G Watt (r.watt@ucl.ac.uk)

Version: 4 Date: 16 April 2011

Author's response to reviews:

Thank you very much for accepting this manuscript.
We have made the formatting changes as requested.
Kind regards,
Ali Golkari